%------------------------------------------------------------------------------
% Here please write the date of submission of paper or its revisions: 30 Apr 2018 
%------------------------------------------------------------------------------
%
\documentclass[12pt, reqno]{amsart}
\usepackage{amsmath, amsthm, amscd, amsfonts, amssymb, graphicx, color}
\usepackage[bookmarksnumbered, colorlinks, plainpages]{hyperref}
\hypersetup{colorlinks=true,linkcolor=red, anchorcolor=green, citecolor=cyan, urlcolor=red, filecolor=magenta, pdftoolbar=true}

\textheight 22.5truecm \textwidth 14.5truecm
\setlength{\oddsidemargin}{0.35in}\setlength{\evensidemargin}{0.35in}

\setlength{\topmargin}{-.5cm}

\newtheorem{theorem}{Theorem}[section]
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{proposition}[theorem]{Proposition}
\newtheorem{corollary}[theorem]{Corollary}
\theoremstyle{definition}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{example}[theorem]{Example}
\newtheorem{exercise}[theorem]{Exercise}
\newtheorem{conclusion}[theorem]{Conclusion}
\newtheorem{conjecture}[theorem]{Conjecture}
\newtheorem{criterion}[theorem]{Criterion}
\newtheorem{summary}[theorem]{Summary}
\newtheorem{axiom}[theorem]{Axiom}
\newtheorem{problem}[theorem]{Problem}
\theoremstyle{remark}
\newtheorem{remark}[theorem]{Remark}
\numberwithin{equation}{section}

\begin{document}
\begin{thebibliography}{99}

\bibitem{1} J. Bourgain and O. I. Reinov, \textit{On the approximation properties for the space $H^{\infty}$}, Math. Nachr. \textbf{122} (1985), 19--27.
 
\bibitem{2} P. G. Casazza, C. L. Garc\'{\i}a and W. B. Johnson, \textit{An example of an asymptotically Hilbertian space which fails the approximation property}, Proc. Amer. Math. Soc. \textbf{129} (2001), no. 10, 3017--3024.
 
\bibitem{3} A. M. Davie, \textit{The approximation problem for Banach spaces}, Bull. London Math. Soc. \textbf{5} (1973), 261--266.
 
\bibitem{4} J. Dieudonn\'e and L. Schwartz, \textit{La dualit\'e dans les espaces $(\mathcal F)$ et $\mathcal{LF}$}, Annales Inst. Fourier \textbf{1} (1949), 61--101.
 
\bibitem{5} T. Figiel and W. B. Johnson, \textit{The approximation property does not imply the bounded approximation property}, Proc. Amer. Math. Soc. \textbf{41} (1973), 197--200.

\bibitem{6} A. Grothendieck, \textit{Produits tensoriels topologiques et espaces nucl\'eaires}, Mem. Amer. Math. Soc. \textbf{16}, 1955.

\bibitem{7} M. J. Kadec and M. G. Snobar, \textit{Certain functionals on the Minkowski compactum} (Russian), Mat. Zametki \textbf{10} (1971), 453--458.

\bibitem{KR} Sten Kaijser and Oleg Reinov, \textit{Approximation properties of the space $H^{\infty}$}, In book: Voronezh Winter Mathematical School "Modern methods of theory of functions and related problems" 2003, Jan 26--Feb 2. Voronezh: VGU (2003), p. 113--114.

\bibitem{8} J. Lindenstrauss, \textit{On James' paper "Separable Conjugate Spaces"}, Israel J. Math. \textbf{9} (1971), 279--284.

\bibitem{9} J. Lindenstrauss and L.Tzafriri, \textit{Classical Banach spaces, vol.1: Sequence spaces}, Berlin-Heidelberg-New York, 1977.

\bibitem{10} E. Oja, \textit{Grothendieck’s nuclear operator theorem revisited with an application to p-null sequences}, Journal of Functional Analysis \text{263} (2012), 2876--2892.

\bibitem{11} E. Oja and O. I. Reinov, \textit{Un contre-exemple \`a une affirmation de A.Grothendieck}, C. R. Acad. Sc. Paris. -- Serie I \textbf{305} (1987), 121--122.

\bibitem{12} A. Pe\l czy{\'n}ski, \textit{Banach spaces of analytic functions and absolutely summing operators}, AMS Regional Conference Series in Mathematics \textbf{30}, Providence, 1977.

\bibitem{13} A. Pietsch, \textit{Operator Ideals}, North Holland, 1980.
 
\bibitem{14} G. Pisier, \textit{Estimations des distances {\`a} un espace euclidien et des constantes de proj{\'e}ction des espaces de Banach de dimensions finie}, Seminaire d'Analyse Fonctionelle 1978-1979, Centre de Math., Ecole Polytech., Paris (1979), expos{\'e} 10, 1--21.
 
\bibitem{15} O. I. Reinov, \textit{Banach spaces without approximation property}, Func. Anal. and Its Appl. \textbf{16} (1982), no. 4, 315--317.

\bibitem{16} O. I. Reinov, \textit{ Approximation properties of order p and the existence of non-p-nuclear operators with p-nuclear second adjoints}, Math. Nachr. \textbf{109} (1982), 125--134.

\bibitem{17} O. I. Reinov, \textit{ A simple proof of two theorems of A. Grothendieck}, Vestn. Leningr. Univ. \textbf{7} (1983), 115--116.

\bibitem{18} Oleg Reinov, \textit{A survey of some results in connection with Grothendieck approximation property}, Math. Nachr. \textbf{119} (1984), 257--264.

\bibitem{19} O. I. Reinov, \textit{Approximation properties $AP_s$ and p-nuclear operators (the case $0 <s\le 1)$}, J. Math. Sciences \textbf{115} (2003), 2243--2250.

\bibitem{20} O. I. Reinov, \textit{On linear operators with $s$-nuclear adjoints, $0<s\le1$}, J. Math. Anal. Appl. \textbf{415} (2014), 816--824.

\bibitem{21} O. Reinov, \textit{Some Remarks on Approximation Properties with Applications}, Ordered Structures and Applications: Positivity VII, 371--394, Trends in Mathematics, 2016.

\bibitem{22} R. A. Ryan, \textit{Introduction to Tensor Products of Banach Spaces}, Springer Monogr. Math., Springer-Verlag, London, 2002.

\bibitem{23} A. Szankowski, \textit{Subspaces without the approximation property}, Isr. J. Math. \textbf{30} (1978) 123--129.

\end{thebibliography}


\setcounter{page}{1}

\title[Some approximation properties and nuclear operators]{Some approximation properties and nuclear operators in spaces of analytical functions}

\author[Sten Kaijser, Oleg I. Reinov]{Sten Kaijser$^1$ \MakeLowercase{and} Oleg I. Reinov$^2$$^{*}$}

\address{$^{1}$Department of Mathematics, Uppsala University, Box 480, S-751 06 Uppsala, Sweden.}
\email{\textcolor[rgb]{0.00,0.00,0.84}{sten@math.uu.se}}

\address{$^{2}$Department of Mathematics and Mechanics, Saint Petersburg State University, St. Petersburg 198504, Russia}
\email{\textcolor[rgb]{0.00,0.00,0.84}{orein51@mail.ru}}

%\dedicatory{This paper is dedicated to Professor ABCD}

%\let\thefootnote\relax\footnote{Copyright 2016 by the Tusi Mathematical Research Group.}

\subjclass[2010]{Primary 46B28; Secondary 46E15.}

\keywords{nuclear operator, tensor product, approximation property, space of bounded analytical functions.}

\thanks{$^{*}$Corresponding author}

\begin{abstract}
We introduce and investigate a new notion of the approximation property $AP_{[c]},$
where $c= (c_n)$ is an arbitrary positive real sequence, tending to infinity. Also,
we study the corresponding notion of $[c]$-nuclear operators in Banach spaces.
Some characterization of the $AP_{[c]}$ in terms of tensor products, as well as
sufficient conditions for a Banach space to have the $AP_{[c]},$ are given.
We give also sufficient conditions for a positive answer to the question:
when it follows from the $[c]$-nuclearity of an adjoint operator the nuclearity of
the operator itself. Obtained results are applied then to the study of properties
of nuclear operators in some spaces of analytical functions.
Many examples are given.
\end{abstract} \maketitle

\section{Introduction and preliminaries}

\subsection{Introduction}\label{S1}
A Banach space $X$ has the approximation property $AP$ if the identity
operator in $X$ can be approximated, in the topology of compact convergence,
by finite rank operators. As was noted by A. Grothendieck \cite[Chap I, Lemma 12, p. 112]{6},
J. Dieudonn\'e and L. Schwartz showed that every compact subset
of a Banach space is contained in the closed convex hull of a sequence,
converging to zero (see \cite[proof of Theorem 5]{4} or \cite[p. 30, Proposition 1.e.2]{9}).
Therefore, the notion of the approximation
property can be define in the following way: the space $X$ has the $AP$ if for every
sequence $(x_n)$ in $X$ with $||x_n||\to 0$ and every $\varepsilon>0$ there exists 
a finite rank operator $R$ in $X$ such that $\sup_n ||x_n-Rx_n||\le\varepsilon.$
It is natural to replace in this definition the condition "for every sequence $(x_n)$ in
$X$ with $||x_n||\to 0$", e. g., by the condition "$\sum ||x_n||^p<\infty$ for some $p\in (0,\infty)$"
and to get corresponding approximation property (say $AP[p]).$
It seems that for the first time it was done in the second author's paper \cite[Lemma 2.1]{18}.
Let us mention that instead of $l_p$-sequences, we can take in the definition any other sets of zero-sequences.

Once a new notion of approximation property is defined, a natural next step is
to study whether some results on the classical approximation property can be
extended to the case of this new notion. As examples, we can consider the characterizations
of the $AP$ in terms of tensor products, some sufficient conditions for a space to have $AP,$
the statements like "$X^*$ has the $AP$ $\implies$ $X$ has the $AP$", the connections between the approximation property
and the properties of so-called nuclear operators, question of whether an operator is nuclear
if its adjoint is nuclear etc.

In this paper, we introduce and investigate approximation properties defined (as above)
by one-point sets of zero-sequences. Namely, suppose $(c_n)$ is a real positive sequence,
tending to $\infty.$ We say that a Banach space $X$ has the approximation property with respect to $(c_n)$
(or the approximation property up to $c_n),$ shortly the $AP_{[c]},$
if for every sequence $(x_n)$ in $X$ with $||x_n||\le 1/c_n$ and every $\varepsilon>0$ there exists
a finite rank operator $R$ in $X$ such that $\sup_n ||x_n-Rx_n||\le\varepsilon.$
Our interest in such properties was inspired by the known fact, that the space $H^\infty$
of bounded analytic functions in the unit disk has the approximation property "up to log" \cite[Theorem 9]{1}.
As a matter of fact, we first had some new properties of nuclear operators, acting from $H^\infty$ or from
the space $L_1/H^1_0$ (predual to $H^\infty)$ (see Sections \ref{S6} and \ref{S7} below)
and then decided to consider, instead of only the sequence $(\log (n+1)),$ also other positive sequences $(c_n)$
with $c_n\to\infty.$

These properties $AP_{[c]}$ are closely connected with a new notion of so-called $[c]$-nuclear operators
For example, a $[\log]$-nuclear operator between Banach spaces $X$ and $Y$ is an operator $T: X\to Y,$
that admits a nuclear representation of type $Tx= \sum_n 1/\log (n+1)\, x'_n(x)\, y_n$ with
$\sum_n ||x'_n||\, ||y_n||<\infty.$ One of the main questions in our study of the classes of $[c]$-nuclear
operators is to give conditions, under which it follows from $[c]$-nuclearity of an adjoint
operator $T^*$ the nuclearity of the operator $T$ itself.

The history of the questions of such a type takes its beginning from a result of A. Grothendieck on the
linear operators with nuclear adjoints. He showed in \cite{6} that if a linear operator $T$ maps a Banach space
$X$ into a Banach space $Y,$ if $T^*$ is nuclear and if the dual space $X^*$ has the approximation property, then the operator
$T$ is nuclear.
This result turned up to be sharp (with respect to the space $X),$ in the sense that there exists
an operator $T$ in a Banach space $X,$ which is not nuclear but has a nuclear adjoint.
This was shown already in 1973 by T. Figiel and W. B. Johnson \cite{5}.
Moreover, in their example, the space $X$ possesses the Grothendieck approximation property.
Later, E. Oja and O. Reinov \cite{11}
gave another sufficient condition for a positive answer to the above question:
If $T$ maps $X$ into $Y,$ $T^*$ is nuclear and $Y^{***}$ has the approximation property, then
the operator $T$ is nuclear. Again, as was shown in the same paper, the condition is essential:
There are a Banach space $Z$ and an operator $T: Z^{**}\to Z$ so that
$Z^{**}$ has a basis, $T^*$ is nuclear but $T$ is not nuclear. Of course, here the space $Z^{***}$
(which is, by the way, separable in the example) does not have the approximation property.

It is easy to see that the nuclearity of an adjoint operator $T^*: Y^*\to X^*$ is equivalent to the nuclearity
of the operator $\pi_YT: X\to Y\to Y^{**},$ where $\pi_Y$ is the natural injection of $Y$ into its second dual
$Y^{**}.$ Thus, the above positive results say that, under some approximation conditions posed on the spaces
$X$ or $Y,$ the space $N(X,Y)$ of all nuclear operators from $X$ to $Y$ possesses a property of "regularity" (recall that an operator ideal $J$
is regular if it follows from $U: X\to Y$ and $\pi_YU\in J(X,Y^{**})$ that $U\in J(X,Y);$ see \cite{13}).
After the Figiel-Johnson example had appeared, the natural questions about
the regularity of such operator ideals as the ideals of $p$-nuclear or $p$-integral (in the sense of A. Pietsch)
operators were posed in 1970s (e. g., by A. Pe\l czy{\'n}ski, A. Pietsch, P. Saphar and others; see \cite{16}).
In a more general setting, a corresponding question on the regularity of the so-called ideals
of $(p,q,r)$-nuclear operators can be found in the book \cite{13}.
Between 1980 and 2017, E. Oja and O. Reinov made several contributions in answering such questions.
We refer here only to a nice paper of E. Oja \cite{10},
where also some corresponding references can be found.

Roughly speaking, we can divide such problems on the regularity (of in different senses nuclear operators)
in three parts: Let $A_1$ be the set of all projective tensor products of Banach spaces,
$A_1^+$ be the set of all tensor products of Banach spaces, equipped with norms,
which are less then the projective tensor norm $||\cdot||_{\land}$ and greater then the operator norm, $A_1^-$
 be the set of all subspaces of all projective tensor products of Banach spaces.
In a natural way, we can define the corresponding classes $\widetilde A_1,$ $\widetilde A_1^+$ and $\widetilde A_1^-$ of operators induced by these sets.
In this setting, we can define a notion of regularity (note that it can be defined in more general cases
of Banach tensor products or quasi-normed tensor products of Banach spaces):
For example, let $X\widetilde\otimes Y$ belong to one of the above sets and $J(X^*,Y)$ be the corresponding set
of operators. We say that $J(X^*,Y)$ is regular if the conditions
$$
U: X^*\to Y,\ \pi_YU \ \text{is generated by an element from}\ X\widetilde\otimes Y^{**}\
$$
imply that $U\in J(X^*, Y)$ and generated by an element from $X\widetilde\otimes Y.$

The cases of the sets $A_1$ and $A_1^+$ (for so-called tensor norms; see \cite{22})
were studied
carefully by E. Oja in \cite{10}.
She showed that if $X$ or $Y^{***}$ has the approximation property, then
every space of operators from $\widetilde A_1$ and $\widetilde A_1^+$ is regular (recall that the corresponding
tensor products in \cite{10} are equipped with tensor norms in the sense of \cite{22}).
Some results, concerning the case of $A_1^-$ can be found in \cite{19} and \cite{20}.

In our paper, we study the case of the sub-set of $A_1^-$ of tensor products, which corresponds
to the approximation properties $AP_{[c]},$ mentioned above.
We will see that some of the approximation properties that were considered and studied in
\cite{19}, \cite{20} and \cite{21} (e.g., $AP_s,$ $AP_{1;p,r}$ $AP_{(pq)})$ as well as the corresponding tensor products
are, essentially, special cases of our considerations.

Shortly about the content of the paper.

In Subsection \ref{S2}, we present some standard notations concerning
Banach spaces, spaces of nuclear operators, tensor products, the approximation property and formulate
a classical result of J. Lindenstrauss from his famous paper \cite{8}, which we will need below.

In Section \ref{S3}, we introduce the notion of the approximation property $AP_{[c]},$ where $c:=(c_n)$
is a positive real sequence, tending to infinity. We also define the notions of so-called $[c]$-projective
tensor products of Banach spaces and corresponding spaces of $[c]$-nuclear operators. The main result here is
Theorem \ref{T3.1}, which gives some characterizations of the $AP_{[c]}$ in terms of tensor products. In the end of the section, we show
that if a dual space $X^*$ has the $AP_{[c]},$ then the space $X$ has the $AP_{[c]}$ too.

In Section \ref{S4}, some sufficient conditions for a Banach space to have the $AP_{[c]}$ are given.
As a consequence, we get an essential generalization of some previous facts about the
approximation properties in subspaces of quotients of $L_p$-spaces. In particular, we show
that every Banach space has the $AP_{[(\sqrt n)]}$ (before it was known that every Banach space has the
approximation property "up to $o(\sqrt n)$", that was a generalization of a famous $2/3$-result of A. Grothendieck).
In the end of the section we present some examples with some assertions. In particular, as a consequence of Example \ref{E4.1},
we obtain the existence of an asymptotically Hilbertian space without the compact
approximation property but with the property $AP_{[\log^{1+\varepsilon}]}$ (see Proposition \ref{P4.2}).

Section \ref{S5} is devoted to a study of operators with $[c]$-nuclear adjoints. We get here an analogue
of the main result of the paper [10], showing that
the set $N_{[c]}(X^*,Y)$ of all $[c]$-nuclear operators from $X^*$ to $Y$ is almost regular (cf. definition of regularity mentioned above)
if either $X$ or $Y^{***}$ has the $AP_{[c]}.$ Namely, if $X$ or $Y^{***}$ has the $AP_{[c]},$
$T: X^*\to Y,$ $\pi_YT$ is generated by an element from $X\widetilde\otimes Y^{**},$
then $T\in N(X^*, Y)$ (i.e., nuclear) and generated by an element from the projective tensor product $X\widehat\otimes Y$ (Theorem \ref{T5.1}).
The main consequence of the theorem is Corollary \ref{C5.1} that gives some sufficient conditions
for an operator to be nuclear if its adjoint is $[c]$-nuclear.

In Section \ref{S6}, some examples are given. These examples show, in particular, that the condition
"either $X$ or $Y^{***}$ has the $AP_{[c]}$" is essential for the results of the previous section,
as well as the conclusion '"$T$ is nuclear" is the best possible (this was known before; see e.g. \cite{20}
and Section \ref{S7} below). After the examples, we present several results on the nuclear operators in some spaces
of analytical functions, for instance: If an operator from $L_1/H^1_0$ has a $[(\log (n+1))]$-nuclear adjoint, then
it is nuclear; if an operator from $H^\infty$ to a Banach space $Y$ is generated by a tensor element
from $L_1\widehat\otimes Y^{**},$ then it is nuclear as an operator from $H^\infty$ to $Y.$

Finally, in Section \ref{S7}, we generalize the notion of the $AP_{[c]}$ to the case where
one considers some subset $\mathcal C_0$ of the set of all positive real sequences,
tending to infinity. We introduce a notion of the approximation property $AP_{\mathcal C_0}$
for a Banach space $X$ (the property means that $X$ has the $AP_{[c]}$ for every sequence $c\in \mathcal C_0).$
Also, we define a corresponding notion of a $\mathcal C_0$-nuclear operator. Examples \ref{E7.1} and \ref{E7.2}
show that in some particular cases we get the notions of some approximation properties and the corresponding
nuclear operators which were studied, for example, in \cite{21}.
We give some generalizations of results from Section \ref{S5} and present some applications.
For example, we show that if $0<s<1,$ $Z$ is either any space of the spaces $A, L_1/H^1_0$ or $H^\infty$
or any of its duals and $T$ is an operator from or into $Z,$ then it follows from $s$-nuclearity of $T^*$
the nuclearity of the operator $T$ itself. The last theorem of the paper (Theorem \ref{T7.3}) is a direct
application of Theorem \ref{T5.1} to the case where one of the Banach spaces under consideration if $H^\infty$.

Let us mention that the results of the paper concerning nuclear operators in the spaces of analytical functions
were partially presented by the authors
at the Voronezh Winter Mathematical School "Modern methods of theory of functions and related problems" (2003, Jan 26--Feb 2, Voronezh, Russia) \cite{KR}.

\subsection{Preliminaries}\label{S2}
All the spaces under considerations $(X,Y, W \dots)$
are Banach, all linear mappings (operators) are continuous; as usual, $X^*, X^{**}, \dots$
are Banach duals (to $X$), and $x', x'', \dots$ (or $y', \dots)$ are the functionals on $X, X^*,\dots$
(or on $Y,\dots).$
If $x\in X, x'\in X^*$ then $\langle x,x'\rangle=\langle x',x\rangle=x'(x).$
$L(X,Y)$ stands for the Banach space of all linear bounded operators from $X$ to $Y;$
$B(X,Y)$ is the Banach space of all continuous bilinear forms on $X\times Y.$
Every Banach space is considered as a subspace of its second dual. If needed,
by $\pi_Y$ we denote the natural isometric injection of $Y$ into $Y^{**}.$

We consider the algebraic tensor product $X\otimes Y$ as the linear subspace of all continuous
finite rank operators from $X^*$ to $Y.$ The projective tensor product $X\widehat\otimes Y$
of the spaces $X$ and $Y$ is the completion of $X\otimes Y$ with respect to the norm
$||z||_{\land}:= \inf \{\sum |\lambda_k|\},$ where the infimum is taken over all finite
representations of $z\in X\otimes Y$ in the form $z= \sum \lambda_k\, x_k\otimes y_k$
with $||x_k||=||y_k||=1.$ Every element $z\in X\widehat\otimes Y$ admits a representation
$z=\sum_{k=1}^\infty \lambda_k x_k\otimes y_k$ such that $\sum |\lambda_k|<\infty$ and
$||x_k||=||y_k||=1.$ We denote by $z^t$ the transposed tensor element from $Y\widehat\otimes X:$
$z^t:= \sum_{k=1}^\infty \lambda_k y_k\otimes x_k.$
 If $X=Y^*,$ then the functional "trace" on the tensor product
$Y^*\widehat\otimes Y$ is well defined by the formula
$\operatorname{trace}\, z:= \sum \lambda_k\, \langle x_k, y_k\rangle.$ The Banach dual
to $X\widehat\otimes Y$ can be identified with the space $L(Y, X^{*})=B(X,Y)$ with duality
given by "trace": for $z\in X\widehat\otimes Y$ and $U\in L(Y, X^{*})$ we put
$\langle U, z\rangle := \operatorname{trace}\, U\circ z=\sum \lambda_k\, \langle x_k, Uy_k\rangle.$

There is a natural map from the tensor product $X\otimes Y$ to $L(X^*,Y),$ that takes elementary
tensors $x\otimes y$ to operators $\langle\pi_Xx,\cdot\rangle y$ of rank one. This map is continuous as a map
from $(X\otimes Y, ||\cdot||_\land)$ to $L(X^*,Y)$ and can be extended to the natural map
$j: X\widehat\otimes Y\to L(X^*,Y).$ We will denote by $N^w(X^*,Y)$ the Banach space of operators
belonging to the image $j(X\widehat\otimes Y)$ of this map (one can identify this space with the
quotient $X\widehat\otimes Y/ \operatorname{Ker}\, j).$
If $X$ is dual to a Banach space, say $W,$ then the corresponding map $j$ can be considered
as a map from $W^*\widehat\otimes Y$ to $L(W,Y).$ We denote by $N(W,Y)$ the image of this map
with a natural norm, induced from the quotient $W^*\widehat\otimes Y/ \operatorname{Ker}\, j.$ The operators from $N(W,Y)$
are called nuclear operators (from $W$ to $Y).$ Thus, in the general case, $N^*(X^*,Y)$ is a subspace
of $N(X^*,Y).$ If $z\in X\widehat\otimes Y$ (or $z\in W^*\widehat\otimes Y),$ then we denote by $\widetilde z$
the corresponding nuclear operator from $X^*$ to $Y$ (or from $W$ to $Y).$

A Banach space $X$ has the approximation property (the $AP),$ if for every $Y$ the canonical
map $j$ is one-to-one. Equivalently, $X$ has the $AP$ if for every Banach space $Y$
the natural map $Y^*\widehat\otimes X\to L(Y,X)$ is one-to-one.
The classical definition of the $AP$ for $X$ is:
A Banach space $X$ has the $AP,$ if
for every compact subset $K$ of $X$ and for any $\varepsilon>0$ one can find a finite rank operator
$R$ in $X$ such that $\sup_{x\in K}\, ||Rx-x||\le \varepsilon.$
See [4] for further information.

We use standard notations for the classical Banach spaces such as $L_p(\mu),$ $C(K),$ $l_p,$ $c_0,$
$l_{pq}$ etc. By $l_{p\infty}^0$ we denote the minimal kernel $l^{min}_{(p, \infty)}$
of $l_{(p, \infty)}$ (see, e. g., \cite[13.9.3 Remark]{13}: A sequence $c=(c_n)$
belongs to $l^{0}_{(p, \infty)}$ if and only if $\lim_n n^{1/p}c_n^*=0,$ where $(c^*_n)$ is a
non-increasing rearrangement of $|c|:= (|c_n|).$
 Finally, $A$ and $H^\infty$ are the disk algebra and the space of bounded analytical functions
respectively. For information about these spaces that is needed, see \cite{12}. Let us mention only that
$A^*= L_1/H^1_0\oplus L$ and $A^{**}= H^{\infty}\oplus L^*,$ where $L$ is an $L_1$-space.

We will need below the following fact from \cite{8} (see Proof of Corollary 1 there):

\begin{lemma}\label{L2.1}
For every separable Banach space $X$ there exist a separable Banach space $Z$ and
a linear homomorphism $ \varphi$ from $ Z^{**}$ onto $ X$ with the kernel
$ Z\subset Z^{**}$ so that the subspace $ \varphi^*(X^*)$ is complemented
in $ Z^{***}$ and, moreover, $Z^{***}\cong \varphi^*(X^*)\oplus Z^*.$
\end{lemma}

\section{Approximation properties $AP_{[c]}$}\label{S3}
It is well known that every compact subset of a Banach space is contained in the closed
convex hull of a sequence converging to 0 (see, e.g., \cite[ p. 112 in Ch.I, Lemma 12]{6},
or \cite[Proposition 1.e.2]{9}. Therefore,
the Grothendieck approximation property for a Banach space $X$ can be defined as follows:
$X$ has the $AP$ if and only if for every sequence $(x_n)_{n=1}^\infty\subset X$ tending to zero, for any $\varepsilon>0$
there exists a finite rank (continuous) operator $R$ in $X$ such that for each $n\in\Bbb N$
one has $||Rx_n-x_n||\le\varepsilon.$

Replacing the set of all $c_0$-sequences in this definition by one fixed sequence, 
we get the following main definition.
Namely, for a positive real sequence $c:= (c_n)$ with $c\to +\infty,$
we define a new notion of the approximation property $AP_{[c]}:$

\begin{definition}\label{D3.1}
A Banach space $X$ has the {\it approximation property up to}\, $c,$ the
$AP_{[c]},$ if for every $\epsilon>0$ and any sequence
$(x_n)$ in $X$ with $||x_n||\le c_n^{-1}$ there exists a finite rank operator $R$ in $X$ such that
$||Rx_n-x_n||\le\epsilon$ for every $n.$
\end{definition}

Let us denote by $\mathcal C$ the set of all such sequences:
$\mathcal C:=\{(c_n):\ c_n\in \Bbb R_{+},\, c_n\to+\infty\}.$

\begin{remark}\label{R3.1}
One can define also the $CAP_{[c]},$ the compact approximation property with respect to $c:$
We need only to change the words "finite rank operator" by "compact operator" in Definition \ref{D3.1}. See Example \ref{E4.1}
and Proposition \ref{P4.2} below.
\end{remark}

Note that $X$ has the classical AP if and only if $X$ has the $AP_{[c]}$ for every $c\in \mathcal C.$

For $c\in \mathcal C,$ let us denote by $X\overset{c}\otimes Y$ a subset of the projective tensor product
$X\widehat\otimes Y,$ consisting of all tensors $z$ such that $z$ admits a representation of type
$$
z=\sum_{n=1}^\infty \mu_n\, x_n\otimes y_n,\ \ \text{where }\ \,
\sum ||x_n||\, ||y_n||<\infty,\, |\mu_k|\le c_k^{-1}\ (k=1, 2, \dots).
$$
Let us note that we can identify (in a sense) $X\overset{c}\otimes Y$ with $Y\overset{c}\otimes X$
 (in a natural way). Also, we can consider $X\overset{c}\otimes Y$ as a subset of $X\overset{c}\otimes Y^{**}$
 (or as a subset of $X^{**}\overset{c}\otimes Y).$ Indeed, $X\overset{c}\otimes Y$ is a subet of
 $X\widehat\otimes Y,$ $X\widehat\otimes Y$ is a subspace of $X\widehat\otimes Y^{**}$ (see \cite[Chap. I, Cor. 3, p. 41]{6})
 and $X\overset{c}\otimes Y^{**}$ is a subset of $X\widehat\otimes Y^{**}.$ Thus the natural map
from $X\overset{c}\otimes Y$ to $X\overset{c}\otimes Y^{**}$ is one-to-one.

On the other hand, we have a natural mapping from $X\widehat\otimes Y$ to $L(X^*,Y)$ (or
to $B(X^*,Y^*)$). Therefore, we can consider also a natural map $j_c$ from $X\overset{c}\otimes Y$ to $L(X^*,Y).$
The image of this map $j_c$ will be denoted by $N^w_{[c]}(X^*,Y).$
In the particular case where the first space is a dual space, say $W^*,$ we get a canonical mapping
$W^*\overset{c}\otimes Y\to L(W,Y).$ The image of this map will be denoted by $N_{[c]}(W,Y)$ and
the operators from the space $N_{[c]}(W,Y)$ will be called $[c]$-nuclear \cite{KR}.

Let us present some characterizations of the $AP_{[c]}$ in terms of tensor products.

\begin{theorem}\label{T3.1}
For $c\in\mathcal C$ and for a Banach space $X,$ the following statements are equivalent:

$1)$\,
$X$ has the $AP_{[c]}.$

$2)$\, For every Banach space $Y$ the natural mapping from $Y\overset{c}\otimes X$ to $B(Y^*,X^*)$
(or to $L(Y^*,X)$) is one-to-one.

$3)$\, For every Banach space $Y$ the natural mapping $Y^*\overset{c}\otimes X\to L(Y,X)$ is one-to-one.

$4)$\, The natural mapping $X^*\overset{c}\otimes X\to L(X,X)$ is one-to-one
(or, what is the same, there exists no tensor element $z\in X^*\overset{c}\otimes X$ with
$\operatorname{trace}\, z=1$ and $\widetilde z=0,$ where $\widetilde z$ is the associated (with $z)$
operator from $X$ to $X$).
\end{theorem}

\begin{proof}
$2)\implies 3)\implies 4)$ --- evident.

$4)\implies 3).$\,
Suppose that there exists a Banach space $Y$ such that the natural map
$Y^* \overset{c}\otimes X\to L(Y,X)$ is not one-to-one. Take an element $z\in Y^*\overset{c}\otimes X$ which is
not zero, but generates a zero operator $\widetilde z: Y\to X.$
Then we can find an operator $U\in L(X,Y^{**})$ so that $\operatorname{trace}\, U\circ z=1.$
If $z=\sum_{k=1}^\infty \lambda_k\, y'_k\otimes x_k$ is a representation of $z$ in $Y^*\overset{c}\otimes X$\,
($\sum ||x'_n||\, ||y_n||<\infty,\, |\lambda_k|\le c_k^{-1},\ k=1, 2, \dots)$, then
$$
1=\operatorname{trace}\, U\circ z= \sum_{k=1}^\infty \lambda_k \langle Ux_k, y'_k\rangle=
\sum_{k=1}^\infty \lambda_k\, \langle x_k, U^*y'_k\rangle
$$
and $\sum_{k=1}^\infty \lambda_k\, U^*y'_k(x) x_k=0$ for every $x\in X.$
Put $x'_k:= \lambda_k U^*y'_k,$ $z_0:= \sum_{k=1}^\infty \lambda_k x'_k\otimes x_k\in X^*\overset{c}\otimes X.$ We have
$$
\operatorname{trace}\, z_0=1,\ \widetilde z_0\neq 0
$$
(by the assumption about $X).$ Consider a 1-dimensional operator $R= x'\otimes x$ in $X$
with the property that $\operatorname{trace}\, R\circ z_0>0.$ Then
$$
0<\operatorname{trace}\, R\circ z_0
= \sum_{k=1}^\infty \lambda_k \langle x'_k, x\rangle \langle x', x_k\rangle
= \sum_{k=1}^\infty \lambda_k\,\langle U^*y'_k, x\rangle \langle x', x_k\rangle
$$
$$= \langle\sum_{k=1}^\infty \lambda_k\, \langle Ux, y'_k\rangle x_k, x'\rangle
= \langle x', \sum_{k=1}^\infty \lambda_k\, U^*y'_k(x) x_k\rangle =0.
$$

$1) \implies 4).$\,
Let $z\in X^*\overset{c}\otimes X$ and $\operatorname{trace}\, z=1.$
Write $z= \sum \lambda_k\, x'_k\otimes x_k,$ where the sequences $(x'_k)$ and
$(x_k)$ are bounded and $(\lambda_k c_k)\in l_1.$
Then
$$
 z=\sum_{k=1}^\infty (\lambda_k c_k\, x'_k)\otimes (c_k^{-1}\, x_k).
$$
Let $\varepsilon>0$ be such that $||(\lambda_k c_k)||_{l_1}\, \sup_k ||x'_k||\cdot \varepsilon< 1/2.$
 By 1), there exists
a finite rank operator $R\in X^*\otimes X$ such that $||R(c_k^{-1} x_k)-c_k^{-1}x_k||\le\varepsilon$
for each $k\in \Bbb N.$ It follows that, for this operator $R,$
$$
|\operatorname{trace}\, (z- R\circ z)| =
|\sum_{k=1}^\infty \langle\lambda_k c_k x'_k, c_k^{-1} x_k- R(c_k^{-1} x_k)\rangle | $$
$$
\le \sum_{k=1}^\infty \lambda_k c_k ||x'_k||\cdot \varepsilon\le
||(\lambda_k c_k)||_{l_1}\, \sup_k ||x'_k||\cdot \varepsilon < 1/2.
$$
Hence, 
$$
|\operatorname{trace}\, R\circ z| \ge 1/2
$$
and therefore $z$ generates a non-zero operator $\widetilde z.$

$3)\implies 2).$
It follows from 3) that for every $Y$ the natural map $Y^{**}\overset{c}\otimes X\to L(Y^*,X)$
is one-to-one. Since $Y\overset{c}\otimes X$ is a subset of $Y^{**}\overset{c}\otimes X,$
we get 2).

$4) \implies 1).$
Suppose that $X$ does not have the $AP_{[c]}$
Then there is a sequence $(x_n)$ such that $||x_n||\le c_n^{-1}$\, $(n=1, 2, \dots)$ and
there exists an $\varepsilon>0$ with the property that for any finite rank operator
$R\in X^*\otimes X$ the inequality $\sup_n ||Rx_n-x_n||>\varepsilon$ is valid.
Consider the space $C_0(K; X)$ for the compact set $K:= \{x_n\}_{n=1}^\infty \cup \{0\}.$
Every operator $U$ in $X$ can be considered as a continuous function on $K$
with values in $X$ by setting $f_U(k):= U(k)$ for $k\in K.$
In particular, for the identity map $\operatorname{id}$ in $X$ and for any $R\in X^*\otimes X$ we have
$$
 ||f_{\operatorname{id}}-f_R||_{C_0(K;X)}\ge\varepsilon.
$$
The subset $\mathcal R:=\overline{\{f_R:\ R\in X^*\otimes X\}}^{C_0(K;X)}$ of $C_0(K; X)$
is a closed linear subspace in $C_0(K; X).$ So, there exists an $X^*$-valued measure
$\mu=(x'_k)_{k=1}^\infty\in C_0^*(K; X)= l_1(\{x_n\}_{n=1}^\infty)\cup \{0\}; X)$ such that
$\mu|_{\mathcal R}=0$ and $\mu(f_{id})=1.$
In other words, we can find a sequence $(x'_k)$ with $\sum_{k=1}^\infty ||x'_k||<\infty$
such that $\sum_{k=1}^\infty \langle x'_k, x_k\rangle =1$ and $\sum_{k=1}^\infty \langle x'_k, Rx_k\rangle =0$
for any $R\in X^*\otimes X.$

Define a tensor element $z\in X^*\widehat\otimes X$ by $z:= \sum_{k=1}^\infty x'_k\otimes x_k.$
Since $||x_n||\le c_n^{-1}$ for $(n=1, 2, \dots)$ and $(x'_k)\in l_1(X^*),$ we get that
$z\in X^*\overset{c}\otimes X,$ $\operatorname{trace}\, z=\sum_{k=1}^\infty \langle x'_k, x_k\rangle =1$ and
$\operatorname{trace}\, R\circ z=0$ for every $R\in X^*\otimes X.$ This means that
the condition 4) is not fulfilled.
\end{proof}

We give here only one of the natural consequences of Theorem \ref{T3.1}:

\begin{theorem}\label{T3.2}
If the dual space $X^*$ has the $AP_{[c]},$ then
$X$ has the $AP_{[c]}$ too.
\end{theorem}

\begin{proof}
We use Theorem \ref{T3.1}. As it is known \cite[Chap. I, Cor. 3, p. 41]{6}, the projective tensor product
$X^*\widehat\otimes X$ is a subspace of the tensor product $X^*\widehat\otimes X^{**}.$
The tensor product $X^*\overset{c}\otimes X $ is a linear subspace of $X^*\widehat\otimes X,$
as well as $X^*\overset{c}\otimes X^{**}$ is a linear subspace of $X^*\widehat\otimes X^{**}.$
Therefore, the natural map $X^*\overset{c}\otimes X \to X^*\overset{c}\otimes X^{**}$
is one-to-one. Now if $X^*$ has the $AP_{[c]},$ then the canonical map
$X^{**}\overset{c}\otimes X^* \to L(X^*,X^*)$ is one-to-one.
Since we can identify the tensor product $X^{**}\overset{c}\otimes X^*$ with
the tensor product $X^{*}\overset{c}\otimes X^{**},$ it follows that
the natural map $X^*\overset{c}\otimes X \to L(X,X)$ is one-to-one.
Thus, if $X^*$ has the $AP_{[c]},$ then $X$ has the $AP_{[c]}$ too.
\end{proof}

.\begin{remark}\label{R3.2}
The converse, generally, is not true: It is well-known that there are Banach spaces with the AP,
whose duals do not have the AP. Hence, if $X$ is one of such spaces, then there exists a sequence
$(x'_n)\subset X^*,$ tending to zero, so that $X^*$ does not have the $AP_{[c]},$ where $c= (||x'_n||^{-1}).$
\end{remark}

\section{Sufficient conditions for a Banach space to have the $AP_{[c]}$}\label{S4}

We give here some examples which are interesting for our notes.
To state them, we formulate and prove the following statement.

\begin{proposition}\label{P4.1}
Let $c\in \mathcal C$ be a non-decreasing sequence.
For a Banach space $Y,$ suppose that

$(\mathcal C)$\ there exist a number sequence $(m_n),$ $m_n\to +\infty,$ and
a positive constant $d$ such that for every natural number $n,$
for every $\varepsilon>0$ and for every subspace $E$ of $Y$ with $\operatorname{dim}\, E\le m_n$
there exists a finite rank operator $R$ in $Y$ so that
$||R||\le d c_{m_n}$ and $||R|_E- {id}_E||_{L(E,Y)}\le \varepsilon.$

Then $Y\in AP_{[c]}.$
\end{proposition}

\begin{proof}
Suppose that there is an element $z\in Y^*\overset{c}\otimes Y$ such that
$ \operatorname{trace}\, z=a>0,$ but $\tilde{z}=0.$ Consider a representation of $z$ of the kind
$$
z=\sum_{k=1}^\infty c_k^{-1}\, y'_k\otimes y_k,
$$
where $\sum_{k=1}^\infty ||y'_k||\, ||y_k||< +\infty.$
Take a decreasing sequence $(b_n)\in c_o$ such that $0\le b_n\le1$ for all $n$ and still
$\sum_{k=1}^\infty b_n^{-1}\, ||y'_k||\, ||y_k||< +\infty.$

Fix a natural number $N,$ large enough, such that for all $m\ge N$
$$
\sum_{k=1}^m c_k^{-1}\, \langle y'_k, y_k\rangle \ge a/2\ \
\text{and }\ \, d\, b_m \sum_{k=m+1}^\infty b_k^{-1}||y'_k|| \, ||y_k||\le a/8.
$$
Fix an $m=m_n, m>N,$ put $E:= {\operatorname{span}} \{y_k\}_{k=1}^m,$
and apply given conditions $(\mathcal C)$
to find a corresponding operator $R\in Y^*\otimes Y$ for $n=m$ and
$\varepsilon= \dfrac{a/4}{\sum_{k=1}^m c_k^{-1}\, ||y'_k||}.$

By our assumption, $ \operatorname{trace}\, R\circ z=0.$
From this, we get
$$
0 = \sum_{k=1}^m c_k^{-1}\, \langle y'_k, Ry_k\rangle +
\sum_{k=m+1}^\infty c_k^{-1}\, \langle y'_k, Ry_k\rangle.
$$
For the first sum:
$$ 
\sum_{k=1}^m c_k^{-1}\, \langle y'_k, Ry_k\rangle \ge
\sum_{k=1}^m c_k^{-1}\, \langle y'_k, y_k\rangle -
\big| \sum_{k=1}^m c_k^{-1}\, \langle y'_k, y_k- Ry_k\rangle \big|$$
$$
\ge a/2 - \sum_{k=1}^m c_k^{-1}\, ||y'_k||\, ||y_k- Ry_k|| \ge
a/2 - \sum_{k=1}^m c_k^{-1}\, ||y'_k||\, \sup_{1\le j\le m}||y_j- Ry_j||
\ge a/2 - a/4=a/4.
$$
For the second sum:
$$
\big| \sum_{k=m+1}^\infty c_k^{-1} \langle y'_k, Ry_k\rangle \big|\le
\sum_{k=m+1}^\infty c_k^{-1}b_k\, b_k^{-1}||y'_k|| \, ||Ry_k||$$
$$
\le c_m^{-1} b_m\, ||R||\, \sum_{k=m+1}^\infty b_k^{-1}||y'_k|| \, ||y_k||
\le d \, c_m^{-1} b_m\, c_m\, \sum_{k=m+1}^\infty b_k^{-1}||y'_k|| \, ||y_k||
=: d_m,
$$
where $0\le d_m \le a/8.$

Now, from the last three relations, we obtain:
$0 \ge a/4 - d_m.$ A contradiction.
\end{proof}

Let us consider some consequences of the proposition \ref{P4.1}.

\begin{corollary}\label{C4.1}
Let $c\in \mathcal C$ be a non-decreasing sequence. For a Banach space $Y,$ suppose that
there exists a constant $d>0$ such that
for every natural number $n$ and for every $n$-dimensional subspace $E$ of $Y$
there exists a finite rank operator $R$ in $Y$ so that
$||R||\le d\, c_n$ and $R|_E= {id}_E.$
Then $Y\in AP_{[c]}.$
\end{corollary}

\begin{corollary}\label{C4.2}
Let $c\in \mathcal C$ be a non-decreasing sequence. For a Banach space $Y,$ suppose that
there exists a constant $d>0$ such that
for every natural number $n$ and for every $n$-dimensional subspace $E$ of $Y$
there exists a finite dimensional subspace $F$ of $Y,$ containing $E$
and $dc_n$-complemented in $Y.$
Then $Y\in AP_{[c]}.$
\end{corollary}

\begin{corollary}\label{C4.3}
Let $c\in \mathcal C$ be a non-decreasing sequence. For a Banach space $Y,$ suppose that
there exists a constant $d>0$ such that
for every natural number $n$ every $n$-dimensional subspace $E$ of $Y$
is $d c_n$-complemented in $Y.$
Then $Y\in AP_{[c]}.$ Moreover, every subspace of the space $Y$
has the $AP_{[c]}.$
\end{corollary}

It is well-known that for each natural number $n$ every $n$-dimensional subspace $E$ of any Banach space $X$
is $\sqrt n$-complemented, i.e. there exists a continuous linear projector $P$ from $X$ onto $E$ with $||P||\le \sqrt n$
(see \cite{7}). Taking in Corollary \ref{C4.3} $c= (\sqrt n),$ we get

\begin{corollary}\label{C4.4}
For any Banach space $X,$ for every $\epsilon>0$ and any sequence
$(x_n)$ in $X$ with $||x_n||\le 1/\sqrt n$ there exists a finite rank operator $R$ in $X$ such that
$||Rx_n-x_n||\le\epsilon$ for every $n.$
\end{corollary}

More generally, let $X$ be a subspace of a quotient of an $L_p$-space $(1\le p\le\infty).$
There is a constant $C(p)$ such that if $n\in\mathbb N$ and $E$ is an $n$-dimensional subspace
of the space $X,$ then there exists a projector $P$ from $X$ onto $E$ with $||P||\le C(p)\, n^{|1/2-1/p|}$
(this follows from \cite[Theorem 4.1 and its Corollaries]{14}). Therefore, we get from Corollary \ref{C4.3}:

\begin{corollary}\label{C4.5}
Let $1\le p\le\infty$ and $X$ be a subspace of a quotient of an $L_p$-space.
For every $\epsilon>0$ and any sequence
$(x_n)$ in $X$ with $||x_n||\le n^{-|1/2-1/p|},$ there exists a finite rank operator $R$ in $X$ such that
$||Rx_n-x_n||\le\epsilon$ for every $n.$
\end{corollary}

\begin{remark}\label{R4.1}
Corollary \ref{C4.4} generalizes a theorem of A. Grothendieck about the property $AP_{2/3}$ (see \cite{6}, \cite{17} or \cite{19}).
Moreover, Corollaries \ref{C4.4} and \ref{C4.5} are generalizations of the corresponding facts, mentioned in
[21, Section 1], where the same conclusions were made for sequences $(x_n)$
with $||x_n||= o(1/\sqrt n)$ or with $||x_n||= o(n^{-|1/2-1/p|})$ respectively. Also, we have
the following generalization of the assertion $(***)'$ in that paper (where the Lorentz spaces
$l_{q,\infty}^0(X)$ were considered):
Given $\alpha\in [0,1/2]$ and a Banach space $X$ with the property that every
finite dimensional subspace $F$ of $X$ is contained in a finite dimensional subspace $E\subset X,$
which in turn is $C\, (\operatorname{dim} F)^\alpha$-complemented in $X,$
we have

$(***)''$\, for every sequence $(x_n)\in l_{q,\infty}(X),$ where $1/q=\alpha,$
for any $\varepsilon>0$ there is a finite rank operator $R$ in $X$ so that
$\sup_n ||Rx_n-x_n||\le\varepsilon.$

Taking into account Theorem \ref{T3.1}, we can reformulate this statement and Corollaries \ref{C4.4} and \ref{C4.5} in terms
of tensor products:
Given $\alpha\in [0,1/2]$ and a Banach space $X$ with the property that every
finite dimensional subspace $F$ of $X$ is contained in a finite dimensional subspace $E\subset X,$
which in turn is $C\, (\operatorname{dim} F)^\alpha$-complemented in $X,$
for any Banach space $Y$ the natural mapping $Y\overset{{}\, n^\alpha}\otimes X\to L(Y^*,X)$ is one-to-one.
Thus, if $1\le p\le\infty, \alpha=|1/2-1/p|$ and $X$ is a subspace of a quotient of an $L_p$-space, then for any $Y$
the natural mapping $Y\overset{{}\, n^{\alpha}}\otimes X\to L(Y^*,X)$ is one-to-one.
In particular, {for any Banach spaces $X$ and $Y$ 
the natural mapping $Y\overset{\sqrt n}\otimes X\to L(Y^*,X)$ is one-to-one.}
Let us mention that the last two statements, as well as $(***)'',$ are optimal in the scale of Lorentz sequence spaces
(this follows, e.g., from Examples, given in \cite{20}).
\end{remark}

\begin{example}\label{E4.1}
In \cite{15} the second author constructed, for every $\varepsilon>0,$ a separable reflexive
space $X_\varepsilon$ with the following properties:
The space $ X_\varepsilon$ does not possess the approximation property.
There exists a constant $ C_\varepsilon>0$ such that
if $ E$ is an $ n$-dimensional subspace of $ X_\varepsilon,$ then
$1)$\, $ d(E, l^2_n)\le C_\varepsilon \log^{1+\varepsilon} n$ and
$2)$\,
$E$\ is $C_\varepsilon\log^{1+\varepsilon} n$-complemented in $ X_\varepsilon.$
Fix $\varepsilon>0.$ By Corollary \ref{C4.3}, the space $X_\varepsilon$ and all of its subspaces
have the $AP_{[\log^{1+\varepsilon}]},$ where $\log^{1+\varepsilon}:=(\log^{1+\varepsilon} (n+1))$
(but, as was said, does not have the $AP).$
Note that it was used in \cite{15} a construction of the space $X_\varepsilon$ from \cite{23}.
Therefore, this space can be taken in such a way that a "bad compact set" $K\subset X$
(in the definition of the $CAP)$ possesses the properties described in \cite{23}.
Therefore, $ X_\varepsilon$ does not possess the $CAP.$
Moreover, this space is an asymptotically Hilbertian space
as follows from the construction in \cite{15}.
\end{example}

\begin{example}\label{E4.2}
One can find a Banach space $W$ such that
$W$ has a Schauder basis and $W^*$ does not have the $AP$ but
has, e.g., the $AP_{[\log^{2}]},$
where $\log^2=(\log^2 (n+1)).$
Indeed, let $X_\varepsilon$ be a separable reflexive Banach space
without the $AP,$ possessing the property $AP_{[\log^{1+\varepsilon}]}$ from Example \ref{E4.1}.
Let $ Z$ be a separable space such that $ Z^{**}$ has a basis
and there exists a linear homomorphism $ \varphi$ from $ Z^{**}$
onto $X_\varepsilon^*$
so that the subspace $ \varphi^*(X_\varepsilon)$ is complemented
in $ Z^{***}$ and, moreover,
$Z^{***}\cong \varphi^*(X_\varepsilon)\oplus Z^*$
(see Lemma \ref{L2.1}).
Put $W:= Z^{**}.$ This (second dual) space $W$ has a Schauder basis
and its dual $W^*$ does not have the $AP,$ but has the $AP_{[\log^{1+\varepsilon}]}.$
We can take $\varepsilon=1$ to get the desired example with $\log^2.$.
\end{example}

These examples \ref{E4.1} and \ref{E4.2} (the space $X_\varepsilon)$ seems to be quite interesting, since, as far as we know, this is the first
example (mentioned in the literature) of an asymptotically Hilbertian space
without the CAP. Let us formulate the result as

\begin{proposition}\label{P4.2}
There exists an asymptotically Hilbertian space without the compact
approximation property. Moreover, this space can be chosen in such a way that
it has the property $AP_{[\log^{1+\varepsilon}]},$ but does not have the $CAP.$
\end{proposition}

Recall that the first example of an asymptotically Hilbertian space without the $AP$
was constructed (by O. Reinov) in 1982 \cite{15}, where A. Szankowski's results were used
(let us note that in that time there was not yet such notion as "asymptotically Hilbertian space").
Later, in 2000, by applying Per Enflo's example in a version of A.~M. ~Davie \cite{3},
P. G. Casazza, C. L. Garc\'{\i}a and W. B. Johnson \cite{2}
gave another example of an asymptotically Hilbertian space which fails the approximation property.
In \cite{20} O. Reinov got another example by using a construction from \cite{13}.

\section{Operators with $[c]$-nuclear adjoints}\label{S5}
The following theorem is one of the main results of the paper and is
an analogue of the main result of the paper \cite{10} (see Introduction above).
Let us mention that the condition
"either $X$ or $Y^{***}$ has the $AP_{[c]}$" is essential in the theorem,
as well as the conclusion '"$T$ is nuclear" is the best possible (see Section \ref{S7} below).
Notations are as above (before Theorem \ref{T3.1}).

\begin{theorem}\label{T5.1}
Let $c\in\mathcal C,$ $ z\in X\overset{c}\otimes Y^{**}),$ $T\in L(X^*,Y)$ be such that $\pi_Y T= \widetilde z\in N^w_{[c]}(X^*, Y^{**}).$
If either
$X\in \,AP_{[c]}\ $ or $\, Y^{***}\in \,AP_{[c]},$ then $T\in N^w(X^*,Y).$
In other words, under these conditions on the spaces involved,
from the $ [c]$-nuclearity of the conjugate (weak${}^*$-to-weak continuous) operator $ T^*: Y^*\to X$ it follows
that the operator $ T$ belongs to the space $ N^w(X^*, Y)$ \, {\rm (in particular, is nuclear)}.
\end{theorem}

\begin{proof}
Suppose there exists a weak${}^*$-to-weak continuous operator $ T\in L(X^*,Y)$ such that
$ T\notin N^w(X^*,Y),$ but $ \pi_Y\,T\in N^w_{[c]}(X^*,Y^{**}).$
Since either $ X$ or $ Y^{**}$ has the $ AP_{[c]}$ (see Theorem \ref{T3.2}),
$ N_{[c]}(Y^*, X)= Y^{**}\overset{c}\otimes X \, ( = X\overset{c}\otimes Y^{**}).$
Therefore the operator $ \pi_Y\,T$ can be identified with the
tensor element
$z\in X\overset{c}\otimes Y^{**}\subset X\widehat\otimes Y^{**};$
in addition, by the choice of $ T,$
$z\notin X\widehat\otimes Y$ \, (the space $X\widehat\otimes Y$
is considered as a closed subspace of the space
$X\widehat\otimes Y^{**}$\!).
Hence there is an operator
$U\in L(Y^{**},X^{*})=\left( X\widehat\otimes Y^{**}\right)^*$
with the properties that
$\operatorname{trace}\, U\circ z=\operatorname{trace}\, \left(z^t\circ \left( U^*|_{X}\right) \right)=1$ and
$\operatorname{trace}\, U\circ \pi_Y\circ u=0$ for each
$u\in X\widehat\otimes Y.$
From the last it follows that, in particular, $ U\pi_Y=0$ and
$\pi_Y^*\,U^*|_{X}=0.$
In fact, if $x\in X$ and $ y\in Y,$ then
$$ \langle U\pi_Y\,y,x\rangle = \langle y, \pi_Y^*\,U^*|_{X}x\rangle
= \operatorname{trace}\, U\circ (x\otimes \pi_Y(y))=0.
$$
Evidently, the tensor element
$ U\circ z\in X\overset{c}\otimes X^{*}$ induces the operator
$ U\pi_Y T,$ which is identically equal to zero.

If $ X\in AP_{[c]}$ then
$X\overset{c}\otimes X^{*}= N^w_{[c]}(X^*,X^{*})$
and, therefore, this tensor element is zero what contradicts
to the equality $\operatorname{trace}\, U\circ z=1.$

Let now $ Y^{***}\in AP_s.$ In this case
$$ V:= \left( U^*|_{X}\right)\circ T^*\circ \pi_Y^*: \
Y^{***}\to Y^* \to X\to Y^{***}
$$
uniquely determines a tensor element
$ z_0$ from the $[c]$-projective tensor product
$ Y^{****}\overset{c}{\otimes}_s Y^{***}.$
Let us take any representation $ z=\sum \mu_n\, x_n\otimes y''_n$ for $ z$
as an element of the space $ X\overset{c}{\otimes}_s Y^{**}.$
Denoting for the brevity the operator $ U^*|_{X}$ by $ U_*,$
we obtain:
$$
Vy'''=U_*\, \left( T^*\pi_Y^*\,y'''\right) =
U_*\, \left( (\sum \mu_n\, y''_n\otimes x_n) \,\pi_Y^*\, y'''\right) $$
$$
=U_*\, \left( \sum \mu_n\, \langle y''_n, \pi_Y^*\, y'''\rangle \,x_n \right) =
\sum \mu_n\, \langle \pi_Y^{**}y''_n, y'''\rangle \,U_* x_n.
$$

So, the operator $ V$ (or the element $ z_0$) has in the space
$ Y^{****}\overset{c}\otimes_s Y^{***}$ the representation
$$ V= \sum \mu_n\, \pi_Y^{**}(y''_n)\otimes U_* (x_n).
$$
Therefore,
$$ \operatorname{trace}\, z_0=\operatorname{trace}\, V= \sum \mu_n\, \langle \pi_Y^{**}(y''_n), U_* (x_n)\rangle =
\sum \mu_n\, \langle y''_n, \pi_Y^*\,U_* x_n\rangle = \sum 0=0
$$
(since $\pi_Y^*\,U_*=0;$ see above).

On the other hand,
$$ Vy'''= U_* \left( \pi_Y T\right)^* y'''= U_*\circ z^t (y''')=
U_*\, \left( \sum \mu_n\, \langle y''_n, y'''\rangle \, x_n\right)=
\sum \mu_n\, \langle y''_n, y'''\rangle \, U_* x_n,
$$
whence $ V=\sum \mu_n\, y''_n\otimes U_*(x_n).$ Therefore
$$ \operatorname{trace}\, z_0=\operatorname{trace}\, V= \sum \mu_n\, \langle y''_n, U_* x_n\rangle = \sum \langle Uy''_n, x_n\rangle
= \operatorname{trace}\, U\circ z=1.$$
A contradiction.
\end{proof}

\begin{corollary}\label{C5.1}
Let $S\in L(X,Y)$ and $S^*\in N_{[c]}(Y^*,X^*).$
If either $X^*\in \,AP_{[c]}\ $ or $\, Y^{***}\in \,AP_{[c]},$ then $S\in N(X,Y).$
\end{corollary}

\begin{proof}
Suppose that the conditions are fulfilled.
Let $S^*= \sum \mu_n\, y''_n\otimes x'_n$ be a representation of $S^*$ in
$N_{[c]}(Y^*,X^*)= Y^{**}\overset{c}\otimes X^*$ (the equality holds by the conditions).

Consider $S^{**}$ as an operator $T$ from $X^{**}$ to $Y:$
$$
Tx'':= \sum \mu_n\, \langle x'_n, x''\rangle y''_n\in \pi_Y(Y)\subset Y^{**},
$$
identifying $Y$ with $\pi_Y(Y)$ in a natural way.
We are in conditions of Theorem \ref{T5.1}. By this theorem, $T\in N^w(X^{**},Y),$
i.e. $T$ admits a nuclear representation
$T=\sum \bar x'_n\otimes y_n$ (with $\sum ||\bar x'_n||\, ||y_n||<\infty).$
But $T|_Xx=Sx$ for all $x\in X.$
\end{proof}

\begin{corollary}\label{C5.2}
Under the conditions of Corollary \ref{C5.1}, if the space $Y$ has the $AP$ and $X=Y,$ then
the nuclear trace of $S$ is well-defined and equal to $\operatorname{trace}\, S^*,$ i.e. to
$\sum \mu_n\, \langle y''_n, x'_n\rangle$ in notation of the proof of Corollary \ref{C5.1}.
\end{corollary}

\begin{proof}
Now, we have $Y^*\widehat\otimes Y=N(Y,Y)$ and $S= \sum y'_n\otimes y_n$ (with $\sum || y'_n||\, ||y_n||<\infty).$
We can consider the tensor element $\sum y'_n\otimes y_n$ as an element of the tensor product
$Y^{**}\widehat\otimes Y^*$ with the same projective norm. On the other hand, the tensor element
$v:=\sum \mu_n\, y''_n\otimes x'_n$ from the proof of Corollary \ref{C5.1} represents the operator $S^*$
and this tensor element must belong to a subspace $\pi_Y(Y)\widehat\otimes Y^*$ of the space $Y^{**}\widehat\otimes Y^*$
(see the proof of Theorem \ref{T5.1}). This means, in particular, that $\langle id_{Y^*}, v\rangle= \langle id_{Y^*}, \sum y'_n\otimes \pi_Yy_n\rangle,$
i.e. $\sum \mu_n\, \langle y''_n, x'_n\rangle = \sum \langle y'_n, y_n\rangle.$
\end{proof}

\section{Examples and some applications}\label{S6}

\begin{example}\label{E6.1}
Fix $\varepsilon>0$ and consider the separable reflexive space $X_\varepsilon$ from Example \ref{E4.1} above.
Recall that this space (as well as its dual) has the property $AP_{[log^{1+\varepsilon}]}$ but does not
have the $AP.$
Apply Lemma \ref{L2.1} to get a separable Banach space $Z_\varepsilon$ with the property that
$Z_\varepsilon^{***}\cong \varphi^*(X_\varepsilon^*)\oplus Z_\varepsilon^*$ ($\varphi$ is the corresponding
linear homomorphism from $ Z_\varepsilon^{**}$ onto $ X_\varepsilon$ with the kernel $Z_\varepsilon).$
Let $u\in X_\varepsilon^*\widehat\otimes X_\varepsilon$ with $\operatorname{trace}\, u=1$ and $\widetilde u=0,$ and let
$u=\sum x'_n\otimes x_n$ be a representation of the tensor element $u$ in the projective tensor product
(where $\sum ||x'_n||\, ||x_n||<\infty).$
Taking $z''_n\in Z_\varepsilon^{**}$ such that $\varphi z''_n=x_n$ and $||z''_n||\le 2\, ||x_n||$
and putting $z'''_n:= \varphi^* x'_n,$ consider $\bar u:= \sum z'''_n\otimes z''_n,$
a tensor element of the projective tensor product $Z_\varepsilon^{***}\widehat\otimes Z_\varepsilon^{**}.$
It is clear that $\operatorname{trace}\, \bar u=1$ and $\widetilde {\bar u}(Z_\varepsilon^{**})\subset \pi_{Z_\varepsilon}(Z_\varepsilon)\subset Z_\varepsilon^{**}$
(since $\varphi \widetilde {\bar u}=0).$ On the other hand, $\widetilde {\bar u}|_{\pi_{Z_\varepsilon}(Z_\varepsilon)}=0.$ It follows
from this that the operator $\widetilde {\bar u},$ considered as an operator from $Z_\varepsilon^{**}$ to $Z_\varepsilon,$
is not nuclear, but is nuclear as an operator from $Z_\varepsilon^{**}$ to $Z^{**}_\varepsilon.$

On the other hand, if we take any operator $T$ from $Z^{**}$ to a Banach space $Y$
(respectively, from a Banach space $W$ to $Z),$ for which the operator $\pi_Y T$
(respectively, the operator $\pi_Z T)$ has a nuclear representation of the kind
$$
\sum_{n=1}^\infty \frac1{\log^{1+\varepsilon}(n+1)}\, z'''_n\otimes y''_n,
$$
where $\sum ||z'''_n||\, ||y''_n||<\infty$
(respectively, of the kind
$$
\sum_{n=1}^\infty \frac1{\log^{1+\varepsilon}(n+1)}\, w'_n\otimes z''_n,
$$
where $\sum ||w'_n||\, ||z''_n||<\infty),$
then the operator $T$ is nuclear, i.e. has a nuclear representation of the kind
$\sum_{n=1}^\infty \hat z'''_n\otimes y_n$ (respectively, $\sum_{n=1}^\infty \hat w'_n\otimes z_n).$
The last follows from Corollary \ref{C5.1}.
\end{example}

\begin{example}\label{E6.2}
Put $X:= L_1/H^1_0$ and use Lemma \ref{L2.1} to define a separable Banach space $Z$ with
the property that $Z^{***}\cong \varphi^*(X^*)\oplus Z^*\cong H^\infty\oplus Z^*.$
Since the space $Z^*$ has a basis and $H^\infty$ possesses the property $AP_{[{\log(1+n)}]}$ \cite{1},
the space $Z^{***}$ has the property $AP_{[{\log(1+n)}]}$ too. Thus, Corollary \ref{C5.1} can be applied and we get:
For any Banach space $W,$ if $U\in L(W,Z)$ and
$$
\pi_Z Uw=\sum_{n=1}^\infty \frac1{\log{(n+1)}}\, \langle w'_n, w\rangle z''_n,\ w\in W,
$$
where $\sum ||w'_n||\, ||z''_n||<\infty,$ then there exist sequences $\{v'_n\}\subset W^*$ and
$\{z_n\}\subset Z$ with $\sum ||v'_n||\, ||z_n||<\infty$ so that
$$
Uw=\sum_{n=1}^\infty \langle v'_n, w\rangle z_n,\ w\in W.
$$
Let us mention that it is unknown whether we can omit "$1/{\log(n+1)}$" above.
\end{example}

Let us consider the first application of our results to the investigation of properties of nuclear operators
in the spaces of analytic functions.

\begin{theorem}\label{T6.1}
Let $T\in L(L_1/H^1_0, Y).$ If there exist sequences $(g_n)\subset H^\infty$ and $(y''_n)\subset Y^**$ such that
$\sum ||g_n||\, ||y''_n||<\infty$ and $T^*y'=\sum 1/\log(n+1)\, \langle y''_n,y'\rangle g_n$ for all $y'\in Y^*,$
then the operator $T$ is nuclear. Moreover, if $Y=L_1/H^1_0,$ then
the nuclear trace of $T$ is well-defined and equals
$\sum 1/\log(n+1)\, \langle y''_n, g_n\rangle.$
\end{theorem}

\begin{proof}
As we know, the space $H^\infty=(L_1/H^1_0)^*$ has the property $AP_{[(\log(n+1))]}.$ Thus,
the first part follows from Corollary \ref{C5.1}.
In the case where $Y=L_1/H^1_0,$ the trace of $T$ is well-defined since the space $L_1/H^1_0$ has the
$AP.$ The equality follows from Corollary \ref{C5.2}.
\end{proof}

\begin{proposition}\label{P6.1}
Let a linear operator $ T: H^{\infty} \to Y $
be such that there are a sequence of functions
$(g_n)\subset L^1 $ and a bounded sequence $(y''_n)\subset Y^{**},$
for which
$\sum_k \int |g_k | \, dm < \infty$
and
$$\pi_YT (f) = \sum_{k = 1}^{\infty}
\int g_k (t) \, f (t) \, dm (t) \, y''_k.
$$
Then the operator $ T $ is nuclear as an operator,
acting from $ H^{\infty} $ into $ Y.$
\end{proposition}

\begin{proof}
Let $S:= T|_A: A\to Y.$ Then $\pi_YS= (\pi_YT)|_A= \sum g^0_k\otimes y''_k\in L_1/H^1_0\widehat\otimes Y^{**},$
where $g^0_k$ is the image of $g_k$ under the quotient map $L_1\to L_1/H^1_0.$
If $y'\in Y^*,$ then
$$
S^*y'= \sum_{k = 1}^{\infty} \langle y''_k, y'\rangle g^0_k\in L_1/H^1_0 \ \text{ and }\ S^{**}|_{H^\infty}= \pi_YT.
$$
Since $A^*$ has the approximation property and $S^*$ is nuclear, the operator $S$ is nuclear too
(by Grothendieck). Now, if $S=\sum \psi_n\otimes y_n$ is a nuclear representation of $S: A\to Y,$
where $\psi_n\in A^*, y_n\in Y, \sum ||\psi_n||\, ||y_n||<\infty,$
then $\pi_YT= \sum \psi_n|_{H^\infty}\otimes \pi_Yy_n$ (we consider here the elements $\psi_n$
as the functionals on $A^{**}).$ It follows that $T=\sum \psi_n|_{H^\infty}\otimes y_n\in N(H^\infty,Y).$
\end{proof}

Note that it is unknown whether the conclusion of the proposition is true if we suppose just that
the operator $\pi_YT$ is nuclear. The same can be said about the next consequence of Proposition \ref{P6.1}.

\begin{corollary}\label{C6.1}
Let a linear operator $T: H^{\infty} \to A$ be such that there are two sequences of functions
$(g_n)\subset L^1 $ and $(f_n)\subset H^{\infty},$ for which $\sum_k \int |g_k \, dm < \infty,$
$\|f_n \| \le1$ for each $n$ and 
$$T(f) = \sum_{k = 1}^{\infty}
\int g_k (t) \, f (t) \, dm (t) \, f_k.
$$
Then the operator $T$ is nuclear as an operator,
acting from $H^{\infty}$ into the disk-algebra $A.$
\end{corollary}

\begin{proof}
As we know, the space $H^\infty$ is a complemented subspace of the second dual $A^{**}.$
Therefore the result follows directly from Proposition \ref{P6.1}.
\end{proof}

\section{Generalizations and further applications}\label{S7}

Recall that we denote by $\mathcal C$ the set
$\mathcal C:=\{(c_n):\ c_n\in \Bbb R_{+},\, c_n\to+\infty\}.$

Let us consider a fixed subset $\mathcal C_0$ of this set
(for example, we can take the set $\{(c_n):\ c_n\in \Bbb R_{+},\, (c_n^{-1})\in l_2\}.$

\begin{definition}\label{D7.1}
A Banach space $X$ has the property $AP_{\mathcal C_0}$ if it has the property
$AP_{[c]}$ for every $c\in \mathcal C_0.$
\end{definition}

\begin{remark}\label{R7.1}
Let us denote by $X\overset{\mathcal C_0}\otimes Y$ the union $\bigcup_{c\in\mathcal C_0} X\overset{c}\otimes Y.$
It is easy to see that Theorem \ref{T3.1} is valid also for this kind of tensor products, i. e. Theorem \ref{T3.1} holds
if we replace $AP_{[c]}$ by $AP_{\mathcal C_0}$ and $\overset{c}\otimes$ by $\overset{\mathcal C_0}\otimes.$
We omit a formulation of this generalized theorem \ref{T3.1}.
\end{remark}

\begin{example}\label{E7.1}
Let $0<p<\infty$ and $\mathcal C_0= l_p^{-1}:= \{c=(c_n):\ c_n\in \Bbb R_{+},\, (c_n^{-1})\in l_p\}.$
By the main definition \ref{D3.1}, a space $X$ has the $AP_{l_p^{-1}}$ if
for any $c\in l_p^{-1},$ for every $\epsilon>0$ and any sequence
$(x_n)$ in $X$ with $||x_n||\le c_n^{-1}$ there exists a finite rank operator $R$ in $X$ such that
$||Rx_n-x_n||\le\epsilon$ for every $n.$ Or: $X$ has this property if and only if
for every $\epsilon>0$ and any sequence $(x_n)$ in $X$ with $\sum ||x_n||^p<\infty$
there exists a finite rank operator $R$ in $X$ such that
$||Rx_n-x_n||\le\epsilon$ for every $n.$
We can see that this property $AP_{l_p^{-1}}$ is exactly the property $\widetilde{AP}_s= AP_s$
from \cite[Sect. 1-2]{21} (see also \cite{19}, \cite{20}),
if we take $s$ from the equality $1/s=1+1/q.$
\end{example}

\begin{definition}\label{D7.2}
An operator $T\in L(X,Y)$ is said to be $\mathcal C_0$-nuclear, if there is a sequence
$c\in\mathcal C_0$ such that $T\in N_{[c]}(X,Y).$
\end{definition}

It is easy to see that Theorem \ref{T5.1} can be formulated and proved for the general case of
$\mathcal C_0$-nuclear operators. But we consider here only an evident generalization of Corollary \ref{C5.1}:

\begin{theorem}\label{T7.1}
Let $S\in L(X,Y)$ and $S^*$ is $\mathcal C_0$-nuclear.
If either
$X^*$ has the property $AP_{\mathcal C_0}$ or $Y^{***}$ has the property $AP_{\mathcal C_0},$
then $S\in N(X,Y).$
\end{theorem}

\begin{example}\label{E7.2}
Let $p\in (0,\infty],$ $r\in (0,\infty]$ and
consider a tensor product $\widehat\otimes_{1; p,r}$ from \cite[Sec. 3]{21}.
It is defined in the following way: For a couple of Banach spaces $X, Y$ the tensor product
$Y\widehat\otimes_{1; p,r} X$ consists of those elements $z$ of the projective tensor product
$Y\widehat\otimes X$ which admit representations of the type
$$
z=\sum_{k=1}^\infty a_kb_k\, y_k\otimes x_k; \ (y_k) \text{ and } (x_k) \text{ are bounded, }
(a_k)\in l_1,\, (b_k)\in l_{pr}
$$
(recall that in \cite{21} and here one considers $l_{p\infty}^0$ in the case $r=\infty).$
If we put $\mathcal C_0:= l^{-1}_{p,r}:= \{c=(c_n):\ c_n\in \Bbb R_{+},\, (c_n^{-1})\in l_{pr}\},$ 
then it is clear that $Y\widehat\otimes_{1; p,r} X= Y\overset{l^{-1}_{pr}}\otimes X$
and the property $AP_{l^{-1}_{pr}}$ is just the property $AP_{1;p,r}$ from \cite{21}
(recall that here we consider $l_{p\infty}^0$ in the case $r=\infty).$

Also, let $0<s<1$ and $0< u\le \infty,$ or $s=1$ and $0< u\le 1.$
If $1+1/p=1/s$ and $1+1/r=1/u,$ then $AP_{1;p,r}= AP_{(s,u)}$ and $Y\widehat\otimes_{1; p,r} X= Y\widehat\otimes_{(s,u)},$
where the last tensor product is the "Lorentz tensor product", consisting of those tensor elements from $Y\widehat\otimes X$
which admit the representations of type
$$
\sum_{k=1}^\infty \lambda_k\, y_k\otimes x_k,\ \text{where} \ (y_k) \text{ and } (x_k)
\text{ are bounded and }
(\lambda_k)\in l_{su}
$$
(see \cite[Sec 3]{21}).
Thus the analogues of Theorems \ref{T3.1} and \ref{T7.1} are valid for the cases of these tensor products
and approximation properties (cf. Remark \ref{R7.1}).
As for the case where $0<p<\infty$ and $\mathcal C_0= l^{-1}_{p\infty},$ we have: $Y\overset{C_0}\otimes X= Y\overset{(n^{1/p})}\otimes X$
and $AP_{\mathcal C_0}= AP_{[(n^{1/p})]}$
(evidently, $\overset{c_0^{-1}}\otimes=\widehat\otimes$ and $AP_{c_0^{-1}}=AP).$ Recall that every Banach space has the $AP_{[(\sqrt n)]}$
(see Corollary \ref{C4.4}). If $p=\infty,$ then we get one more new result considering $\mathcal C_0=l_{p\infty}$ in Theorem \ref{T7.1}
(otherwise we can apply the "generalized" Theorem \ref{T3.1} to this case; cf. Remark \ref{R7.1}).
\end{example}

Note that a partial case of Theorem \ref{T7.1} (namely, the case considered in Example \ref{E7.1}) was studied
already in \cite{20}.
In this case, the corresponding operators are called $s$-nuclear (recall that $1/s=1+1/p).$
Let us apply Theorem \ref{T7.1} to the particular case of the given spaces of analytic functions.

\begin{theorem}\label{T7.2}
Let $0<s<1$ and $W$ be any of the following Banach spaces: $A, L_1/H^1_0$ or $H^\infty.$
Let $Z$ be either the space $W$ or any of its duals $(W^*$ or $W^{**}$ etc.).
If $Y$ is a Banach space, $T\in L(Z,Y),$ $U\in L(Y,Z),$ $T^*$ and $U^*$ are $s$-nuclear,
then $T\in N(Z,Y)$ and $U\in N(Y,Z).$
\end{theorem}

\begin{proof}
As was shown in \cite[Theorem 1]{1}, the space $H^\infty$ and all of its duals have the property $AP_s$
for any $s\in (0,1).$ Therefore, the same is true for the spaces $A$ and $L_1/H^1_0.$
Now, the assertion of Theorem \ref{T7.2} follows from the considered partial case of Theorem \ref{T7.1}.
\end{proof}

\begin{remark}\label{R7.2}
In the case where $s=1,$ the assertion of Theorem \ref{T7.2} is known to be valid only
if $Z=A$ and $T^*\in N(Y^*, A^*)$ (cf. the proof of Theorem \ref{T6.1}).
\end{remark}

We end the paper with a direct application of Theorem \ref{T5.1} to some nuclear bilinear forms
on the products of type $H^\infty\times Y:$

\begin{theorem}\label{T7.3}
Let $\log:=(\log (n+1))$ and $T\in L((H^\infty)^*, X)$ be such that $T^*(X^*)\subset H^\infty.$
 If there is a tensor element
$z\in X^{**}\overset{\log}\otimes H^\infty$ which generates the operator $T^*,$ then $T$
is a nuclear operator from $(H^\infty)^*$ to $X$ that can be generated by a tensor element
belonging to the projective tensor product $H^\infty\widehat\otimes X.$
\end{theorem}

%%


\bibliographystyle{amsplain}


\end{document}

%------------------------------------------------------------------------------
% End of journal.tex
%------------------------------------------------------------------------------
